



Department of Economics

Athens University of Economics and Business

WORKING PAPER no. 19-2023

From debt arithmetic to fiscal sustainability and fiscal rules: Taking stock

George Economides, Natasha Miaouli and Apostolis Philippopoulos

November 2023

The Working Papers in this series circulate mainly for early presentation and discussion, as well as for the information of the Academic Community and all interested in our current research activity.

The authors assume full responsibility for the accuracy of their paper as well as for the opinions expressed therein.

From debt arithmetic to fiscal sustainability and fiscal rules: Taking stock^{*}

George Economides, Natasha Miaouli[‡] and Apostolis Philippopoulos[§]

November 1, 2023

Abstract

We start by clarifying the role of the interest rate-growth rate differential in debt arithmetic with a numerical example for the Greek economy. In turn, building upon this standard and popular approach to fiscal sustainability, which is based on the intertemporal government budget constraint only, we make a number of methodological points that question its quantitative usefulness. A structural approach is thus needed and this reveals the necessity of fiscal rules according to which fiscal instruments react to public debt imbalances. This naturally enables us to evaluate the EU's fiscal rules and to suggest simple and implementable alternatives. Throughout, we confront our arguments with data from the Euro Area.

JEL classification: E62, H63 **Keywords:** Fiscal policy, public debt dynamics, Euro Area

^{*}This is an expanded version of the note by GE and AP in EconPol Forum (2023). We thank V. Dimakopoulou, G. Hodroyiannis, G. Koliousi, J. Malley and E. Tzavalis for discussions and comments. NM and AP clarify that the views here may not reflect the views of the Hellenic Fiscal Council. Any errors are ours.

[†]Athens University of Economics and Business, CESifo, and Hellenic Open University [‡]Hellenic Fiscal Council

[§]Athens University of Economics and Business, CESifo, Hellenic Observatory at the LSE, and Hellenic Fiscal Council. Corresponding author: Department of Economics, Athens University of Economics and Business, 76 Patission street, Athens 10434, Greece. email: aphil@aueb.gr

1 Introduction

Fiscal sustainability is a necessary condition for macroeconomic stability which, in turn, is a prerequisite for economic growth and the funding of social policies. But how can we judge fiscal sustainability? The popular approach, at least in policy reports and public debates, is based on the inter-temporal government budget constraint (IGBC); see, for example, the European Commission's longterm fiscal sustainability indicators S1 and S2, as well as its recommendations for the public finances of EU countries (European Commission (2023a)). This means that one calculates the primary fiscal balance required to hit a target value for the public debt to GDP ratio after a certain number of time periods. As is known, in this kind of analysis which is also known as public debt arithmetic, the required primary fiscal balance and hence fiscal sustainability depend critically on the comparison between the real interest rate on sovereign bonds and the economy's real growth rate, both of which are assumed to be exogenous variables.

We will therefore start with this standard approach to fiscal sustainability. Using data, for example, from Greece which is the country with the highest public debt to GDP ratio in the EU, we will provide a numerical solution that illustrates the importance of the interest rate-growth rate differential. In particular, we will show that when the growth rate is assumed to exceed the interest rate by one percentage point, the public debt ratio can be brought down from its current level of 171% to 100% in say 35 years from now without any extra fiscal effort, simply by keeping the primary fiscal balance almost balanced on average over time. By contrast, when the interest rate is assumed to exceed the growth rate by one percentage point, other things equal, a primary fiscal surplus of around 3.4% of GDP is required on average in each year for the same goal. A 3.4% primary surplus is rather demanding! This kind of arithmetic also illustrates the importance of public policies that enhance growth and trust for fiscal sustainability (a loss of trust is immediately reflected in an increase in sovereign interest rates).

In turn, building upon the above, we will make four points.

First, we argue that, for countries with official obligations to EU institutions, things are more demanding. In Greece, for example, standard calculations, like the above, do not take into account the country's obligations to EU institutions - specifically, that the outstanding debt to ESM, EFSF, etc, from the three official fiscal bailouts of the previous decade amounting to around 250 billion euros, has been agreed to be repaid by around 2070. Once this is taken into account, the required fiscal surplus is higher than that implied by the standard calculations.

Second, one should be careful how to read the ranking of countries according to the European Commission's long-term fiscal sustainability indicators S1 and S2, because they are sensitive to the assumption that the fiscal situation in the departure year will not change over time. This can contribute to explaining why, for example, Greece is ranked as a low fiscal risk country in the long run, while, Germany is ranked as a medium fiscal risk country in the European Commission's (2023, chapter 3) report.

Third, and more importantly, we stress that debt arithmetic analyses like the above, although conceptually and educationally useful, are less reliable quantitatively because they suffer from the Lucas critique (see also D' Erasmo et al. (2016)). This is because the real interest rate on sovereign bonds, the economy's real growth rate, as well as most items incorporated in the primary fiscal balance (tax revenues for instance), are all endogenous variables depending on a number of factors and policies including the level of public debt and fiscal policy reactions to it. Hence, a structural approach is needed and this rationalizes the use of dynamic general (dis)equilibrium macroeconomic models. Then, a common implication of such models is that dynamic stability and hence fiscal sustainability require debt-contingent fiscal rules according to which fiscal instruments (like public spending items and tax rates) react to the gap between the outstanding public debt and a policy target value; simply, this is a necessary condition to get a solution.

Fourth, the necessity of debt-contingent fiscal reaction functions allows us to contribure to the current debate on the EU's fiscal rules. After evaluating the EU's recently proposed expenditure rule (see European Commission (2023b)), we suggest that, in addition to the public debt gap, fiscal instruments should be contingent on the interest rate-growth rate differential and the gap between the primary fiscal balance from its medium-term objective (these are simply the variables that jointly shape the public debt dynamics).

Throughout the note, we confront our arguments with data from the Euro Area (EA).

Section 2 presents the government budget constraint and solves it depending on the interest rate-growth rate differential. Section 3 provides numerical solutions, or debt arithmetic, using Greek data. Section 4 presents data on real interest rates and growth rates in the EA countries. Section 5 evaluates some of the EC's fiscal sustainability criteria that rely on the government budget constraint. Section 6 adds debt-based rules to debt arithmetic. Section 7 argues for the use of dynamic general (dis)equilibrium models. Section 8 presents evidence of fiscal reaction to public debt in EA countries. Section 9 evaluates the EU's fiscal rules and suggests alternatives. Section 10 concludes.

2 Fiscal sustainability through the lens of the government budget constraint

Most policy reports (see e.g. European Commission (2023a) for a recent example) analyse the issue of fiscal sustainability through the lens of the government budget constraint. We therefore start by presenting the government budget constraint, writing it in terms of GDP and then solving the resulting difference equation for public debt as proportion of GDP in two different ways depending on the interest rate-growth rate differential. This is pretty standard macroeconomics but it will be necessary for the debt arithmetic that follows in the next section. It will also allow us to evaluate some popular indicators of fiscal sustainability as well as to understand what type of policies can reduce the debt ratio.

2.1 Government budget constraint

The within-period government budget constraint is (see e.g. Walsh (2017, chapter 4) and Buiter (2021, chapters 1 and 2) for details):

$$G_t + i_t B_{t-1} \equiv (B_t - B_{t-1}) + T_t + N_t \tag{1}$$

where G_t is total government spending except interest payments, T_t is total tax revenues, B_t is the end-of-period total public debt, N_t is transfers from the CB to its government (if any)¹ and i_t is the nominal interest rate on outstanding one-period bonds, B_{t-1} . All variables are expressed in nominal terms. Notice that for simplicity we assume that bonds have one period maturity.

If we express nominal quantities as shares of nominal GDP, we have:

$$\frac{B_t}{Y_t} \equiv R_t \frac{B_{t-1}}{Y_{t-1}} + \left(\frac{G_t}{Y_t} - \frac{T_t}{Y_t} - \frac{N_t}{Y_t}\right)$$

where $R_t \equiv \frac{1+i_t}{(1+\pi_t)(1+\gamma_t)}$, $\pi_t \equiv \frac{p_t-p_{t-1}}{p_{t-1}}$ is the inflation rate and $\gamma_t \equiv \frac{y_t-y_{t-1}}{y_{t-1}}$ is the growth rate of real GDP. Notice that, approximately, $R_t \equiv \frac{1+i_t}{(1+\pi_t)(1+\gamma_t)} \approx 1+i_t - \pi_t - \gamma_t$, where $i_t - \pi_t$ is the usual definition for the real interest rate.

In a shorter notation, we have:

$$b_t \equiv R_t b_{t-1} + d_t \equiv R_t b_{t-1} - s_t \tag{2}$$

where $b_t \equiv \frac{B_t}{Y_t}$ is the end-of-period public debt to GDP ratio, $b_{t-1} \equiv \frac{B_{t-1}}{Y_{t-1}}$ is the beginning-of-period public debt to GDP ratio and $d_t \equiv \left(\frac{G_t}{Y_t} - \frac{T_t}{Y_t} - \frac{N_t}{Y_t}\right)$ is the primary fiscal deficit as share of GDP or, symmetrically, $s_t \equiv -d_t \equiv \left(\frac{T_t}{Y_t} + \frac{N_t}{Y_t} - \frac{G_t}{Y_t}\right)$ is the primary fiscal surplus.

Therefore, as equation (2) shows, the drivers of the debt ratio over time are the interest rate-growth rate differential, as captured here by R_t , and the primary balance, d_t (or s_t). Techically speaking to the extent that R_t and d_t (or s_t) are assumed to be exogenous and in particular independent of public debt (see below for this assumption), this is a first-order linear difference equation in the path of b_t , whose stability, and hence the conditions for fiscal sustainability, depend crucially on the value of R_t . Typically, we distinguish two cases, $R_t < 1$ and $R_t > 1$ (see e.g. Blanchard et al (1990) and Wickens (2008, chapter 5)).

¹This transfer consists of seigniorage revenue and interest income from the central bank's net assets. See Buiter (2021) for details and numbers; seigniorage revenue can be large at the effective lower bound, i.e. in liquidity traps, but very small in more normal periods.

2.2 Favorable interest rate-growth rate differential

If $R_t < 1$, namely if the interest rate-growth rate differential is favorable, equation (2) is stable. A model that satisfies this condition is said to be stationary. In this case, since the cost of inherited debt steadily declines over time, the government does not need to generate primary surpluses to achieve sustainability. To the extent that the sequence $\{d_t\}$ is bounded, the government can simply roll over its debt, issuing new debt to pay for the interest, without the need to cut spending or raise taxes in the future for the debt to GDP ratio to remain finite (see e.g. Blanchard et al (1990), Wickens (2008, chapter 5.4) and Blanchard (2019)). Note however that, in practice, even if the government can run permanent primary deficits and these deficits can lead to a finite public debt to GDP ratio, there might be fears of default if this finite ratio is considered to be "too" high.² But, even without default fears, a high debt ratio limits the room for fiscal manoeuver and support of the economy in case of downturns in the future.³ These can provide extra arguments for upper limits on the debtto-GDP ratio like those of the Stability and Growth Pact in the EU even when the differential is favorable (see Wickens (2008, chapter 5.4.2) and Blanchard (2019) for the economic intuition behind the stable case).

Since $R_t < 1$, equation (2) can be solved backward.⁴ By repeated substitutions, we get (here, for simplicity, we assume that the exogenous R_t and d_t remain constant over time):

$$b_t \equiv d \sum_{i=0}^{t-1} R^i + R^t b_0$$
 (3)

or in a simpler way:

$$b_t \equiv d\left(\frac{1-R^t}{1-R}\right) + R^t b_0 \tag{4}$$

where b_0 is the initial debt to GDP ratio.

Summing up, when the interest rate-growth rate differential is favorable and the path of primary fiscal deficits is simply bounded, the debt to GDP ratio will remain finite and hence the fiscal stance is sustainable.

Notice that the above relates to the so-called "revised S1 indicator" used for long-term fiscal sustainability analysis by the EC (see European Commission (2023, chapter 3 and Annex A5.3)). In particular, our equations (3)-(4) can be compared to equations (4)-(5) in the EC's Annex A5.3. Details on S1 are provided in Section 5 below.

 $^{^{2}}$ As Jones (2008, chapter 13) points out, there is no magic level of the debt to GDP ratio that triggers such a calamity. The level depends on a number of economic and political fundamentals.

 $^{^3 \}rm See$ e.g. the early papers by Buiter and Kletzer (1992) and Friedman (1992) for the consequences of fiscal deficits and public debt.

⁴We work as in e.g. Sargent (1987, chapter IX) and Azariadis (1993, chapter 2). Appendix A at the end of this note provides algebraic details for the more general case where d_t changes over time.

2.3 Unfavorable interest rate-growth rate differential

If $R_t > 1$, the public debt ratio is not stationary meaning that, given d_t , its path is explosive over time. In this case, there are two ways to restore stability and hence sustainability (see e.g. D'Erasmo et al (2016)).

First, we can introduce a feedback fiscal policy rule according to which a fiscal instrument reacts to outstanding public debt so that the "effective" coefficient on outstanding debt becomes less than one; this is analysed in detail below.

Second, since $R_t > 1$, equation (2) can be solved forward.⁵ By repeated substitutions, we get (here, as above, for simplicity, we assume that R_t and $s_t \equiv -d_t$ remain constant over time):

$$b_{t-1} \equiv \frac{s}{R} \sum_{i=0}^{T} \left(\frac{1}{R}\right)^i + \left(\frac{1}{R}\right)^{T+1} b_{t+T} \tag{5}$$

or in a simpler way:

$$b_{t-1} \equiv \frac{s}{R} \frac{1 - \left(\frac{1}{R}\right)^{T+1}}{1 - \frac{1}{R}} + \left(\frac{1}{R}\right)^{T+1} b_{t+T}$$
(6)

so that current liabilities are equal to the present discounted value (PDV) of expected future fiscal surpluses plus the discounted value of the end-of-horizon debt ratio.

If no other side condition is imposed, we usually assume an infinite time horizon and impose the so-called transversality condition $\lim_{T\to\infty} \left(\frac{1}{R}\right)^{T+1} b_{t+T} = 0$. From an economics point of view, such a condition excludes Ponzi-type games (see e.g. Wickens (2008, p. 100) and European Commission (2023, Annex A5)).⁶ From an algebraic point of view, this condition looks innocent at first sight since $\frac{1}{R} < 1$ is raised to a large number, but this presupposes that the future debt ratio, b_{t+T} , is finite; if, however, this forward-looking variable is thought of as an asset price, then self-fulfilling rational bubbles cannot be excluded so that the expected value of the debt ratio can become explosive over time (see Blanchard and Fischer (1989, chapter 5) for a proof). In what follows, as in most policy papers, we will assume that there is a side condition that ties down the future value of b_{t+T} in the intertemporal government budget constraint (5)-(6).

Summing up, when the interest rate-growth rate differential is unfavorable, a finite debt to GDP ratio (and hence fiscal sustainability) requires not only that the path of primary fiscal balances is bounded as in the favorable case studied above, but also that the PDV of expected future primary fiscal surpluses plus the discounted value of the end-of-horizon debt ratio (where, in the research

⁵We again work as in e.g. Sargent (1987, chapter IX) and Azariadis (1993, chapter 2). Appendix B at the end of this note provides algebraic details for the more general case where d_t changes over time.

 $^{^6 \}rm See$ also Obstfeld and Rogoff (1996, chapter 2) for the intuition of such terminal conditions although in a different setup.

literature, the latter is typically set to zero) are large enough to meet the current liabilities of the government.

Notice that the above relates to the so-called "S2 indicator" used for longterm fiscal sustainability analysis by the EC (see European Commission (2023), chapter 3 and Annex A5.4)). In particular, our equations (5)-(6) can be compared to equations (7)-(9) in the EC's Annex A5.4. Details on S2 are provided in Section 5 below.

3 Numerical solutions and public debt arithmetic

Using the above analytical framework, we will now illustrate the quantitative importance of the interest rate-growth rate differential for the fiscal balance required for fiscal sustainability. That is, using the simple tool of the government budget constraint, we will quantify the required fiscal adjustment under different assumed scenaria regarding the interest rate-growth rate differential as well as the target for the public debt to GDP ratio at some future time. This is known as debt arithmetic. As an example, as said above, we will refer to the case of Greece being the country with the highest public debt to GDP in the EU.

3.1 The interest rate-growth rate differential and its role in the standard debt arithmetic

We start with a scenario of an unfavorable interest rate-growth rate differential. Thus, we work with equation (6). In particular, let us say that the outstanding public debt to GDP ratio is 171% as it was the case in Greece at the end of 2022. Also say that the nominal interest rate is 4%, the inflation rate is 2%, so that the real interest rate is 2%, and that the growth rate of real GDP is 1%(these are the numbers also used by Buiter (2021, chapter 1) for the EA). In other words, we assume R = 1.01 > 1. We also assume a time horizon of say 35 years, i.e. T = 35, at the end of which the public debt ratio is simply set at its starting value, i.e. $b_{t+T} \equiv b_{t-1} = 1.71$. That is, we ask what primary balance is needed to keep the debt ratio after 35 years equal to the debt ratio today (see also Blanchard and Das (2017) for similar exercises). This is a case of debt stabilization. Then, solving equation (6) for s gives $s \approx 0.017$.⁷ In other words, public debt stability requires a primary annual surplus of 1.7% of GDP on average over the coming 35 years. If, on the other hand, we assume that, at the end of the 35 years, the public debt ratio is lower than its starting value as recommended by the EC, say 100% of GDP, then $s \approx 0.034$.⁸ This is a case of debt consolidation. In other words, according to this more ambitious scenario where the end-of-horizon debt is lower than the current one, the average surplus should be 3.4% of GDP. Note that these numbers are close to those reported

⁷That is, s solves $1.71 = \frac{s}{1.01} \frac{1 - \left(\frac{1}{1.01}\right)^{36}}{1 - \frac{1}{1.01}} + \left(\frac{1}{1.01}\right)^{36} 1.71.$ ⁸That is, s solves $1.71 = \frac{s}{1.01} \frac{1 - \left(\frac{1}{1.01}\right)^{36}}{1 - \frac{1}{1.01}} + \left(\frac{1}{1.01}\right)^{36} 1.$

by the EC in its Post Programme Surveillance Report on Greece published in Autumn 2022 (see European Commission (2022b, p. 18)); the latter reports numbers between 1.4% (under a relatively optimistic scenario about the gap between the real interest rate and the growth rate) and 3.1% (under a relatively pessimistic scenario about the same gap).

The above can be compared to a favorable interest rate-growth rate differential. We use the same parameter values as above except that now we set, for example, R = 0.99 < 1. Thus, now we work with equation (4). Focusing on the relatively ambitious case in which the end of period debt is 100% of GDP, solving equation (4) for d gives $d \approx -0.007$ or $s = -d \approx 0.7\%$ of GDP.⁹ This primary surplus of 0.7% is much smaller than 3.4% which was the solution under the adverse differential above, other things equal. A favorable interest rate-growth rate differential erodes the burden of outstanding public debt so that public debt can be brought down without any extra fiscal effort (here by just keeping the primary fiscal balance almost balanced).

In sum, as is well recognized, the interest rate-growth rate differential makes a lot of difference in terms of the fiscal effort needed to achieve fiscal sustainability. This is in particular so in high public debt countries like Greece. At this point, it is also useful to compare the fiscal effort required for fiscal sustainability to the actual data.¹⁰ If big primary fiscal deficits - like those experienced during the global financial crisis, the pandemic crisis and the energy-food crisis - become a normality, then the numbers for fiscal effort that come out from debt arithmetic exercices like the above are quite demanding, even if we assume a favorable interest rate-growth rate differential. The climate crisis and the ageing problem are additional fiscal risks (see e.g. Schuknecht (2022) for a discussion of risks and public finances).

3.2 How inflation erodes the real debt burden - alas, temporarily

Say that b_{t-1} is 193% of GDP, the nominal interest rate on outstanding debt, i_t , is 2%, the inflation rate, π_t , is 10% and the real GDP growth rate, γ_t , is 4%. Let us also assume a primary deficit of around 2% of GDP. These numbers are very close to the actual Greek data in the year 2022. Then, equation (6) above implies that at the end of 2022 the public debt ratio would be around 175%, which is substantially below the starting value of 193% and, actually, is close to the Greek data (see European Commission (2022c)). That is, even with a fiscal deficit, the debt to GDP ratio has decreased over time. In this example, this happens thanks to growth but mainly thanks to inflation which erodes the real burden of outstanding public debt and hence reduces the end-of-period public debt to GDP ratio.

⁹ That is, d solves $1 = d\left(\frac{1-0.99^{35}}{1-0.99}\right) + 0.99^{35}1.71.$

 $^{^{10}}$ For primary fiscal balances in the EA and the EU, see e.g. the Economic Forecasts of the EC over the years. Wyplosz (2014) and Alesina and Passalacqua (2016) summarize historical data for fiscal deficits in OECD countries.

As already mentioned above, this has been one of the classic ways to reduce public debt burdens and hence public debt ratios in the world history of debt (see e.g. Dornbusch and Draghi (1990)). But, as is widely recognized, this is a short-term resolution only to debt stabilization. In addition to the standard redistributive and aggregate costs associated with high inflation, high inflation also means that the government will sooner or later have to make concessions. The latter typically include a mix of rising interest rate premia on long-term bonds, a shift to shorter maturities and the issuance of indexed bonds (of course, all this applies to newly issued bonds). If such things occur, sooner or later, the burden of adjustment will shift to higher taxes and/or spending cuts.

In sum, unexpected inflation can help the public finances but, alas, temporarily only. High inflation and low nominal interest rates just buy time. See also International Monetary Fund (2023, chapter 3) for the undesirable effects of high inflation as a means of reducing debt ratios.

3.3 A more careful debt arithmetic for countries with obligations to EU institutions

The above may be helpful to understand the standard methodology used in policy circles to evaluate debt sustainability but it ignores the extra obligations of some highly indebted countries to EU public institutions. Using again Greece as an example, a large part of the Greek public debt is in the hands of non-market EU institutions (ESM, EFSF, etc) as a result of the three official fiscal bailouts in the 2010s amounting to around 290 billion euros, and the country's obligation is that all this has to be paid back between 2060 and 2070 (by the year 2060 for ESM loans and by 2070 for EFSF loans). In this section, we will add this to the previous analysis. Note that we keep working with the government budget constraint only.

Decomposing the total public debt into that held by private agents/banks and that held by non-market EU institutions, we rewrite (2) as:

$$b_t^p + b_t^{eu} \equiv R_t^p b_{t-1}^p + R_t^{eu} b_{t-1}^{eu} - s_t \tag{7}$$

where the superscripts p and eu refer to public debt owed to private agents/banks and non-market EU institutions respectively and s_t denotes the total primary fiscal surplus (we again assume one period debt maturity for simplicity).

We will work in two steps. In the first step, we will calculate the average over time fiscal surplus needed to pay back the debt to the EU in, say, 35 years. Then, in the next step, we will check what this implies for the other part of the debt (namely, the private one) and hence for the total primary fiscal surplus required for sustainability.

Regarding the fraction of Greek public debt in the hands of non-market EU institutions this has been estimated to be at least 70% of the total Greek public debt (see Dimakopoulou et al (2022)). If we assume that the non-market nominal interest on this part of the debt is 1%, and, as assumed above, inflation is 2% and the growth rate is 1%, and that all of them remain constant over time,

this implies $R_t^{eu} = 0.98 < 1$, which in turn means that the associated difference equation for this part of the debt, $b_t^{eu} \equiv R_t^{eu} b_{t-1}^{eu} - s_t^{eu}$, is dynamically stable. Thus, we can use equations (3)-(4) above. Setting $b_t^{eu} = 0.7x1.71$ and assuming that after 35 years this part of debt is fully repaid, equation (4) implies that, other things equal, this requires an average primary fiscal surplus of $s_t^{eu} \cong 0.023$ or 2.3% over the next 35 years.¹¹ Before we move on, it is worth examining what s_t^{eu} would be in the counter-factual case in which the nominal interest rate on the EU debt were the market one. In particular, we examine what happens if $b_t^{eu} \equiv R_t^{eu} b_{t-1}^{eu} - s_t^{eu}$, where, by using the same numbers as in the previous section, $R_t^{eu} = 1.01 > 1$. In this case, we have to use equation (6) which gives $s_t^{eu} \cong 0.04$ or 4%.¹² This is much higher than 2.3%.

In turn, we plug this value of 0.023 as a constant term (denoted as a) on the expenditure side of equation (B.3) in Appendix B which is a generalization of equation (6) above. Thus, now, the associated difference equation is $b_t^p \equiv$ $R_t^p b_{t-1}^p - s_t^p + 0.023$, where $R_t^p = 1.01 > 1$. As in the previous section, let us first study the less ambitious scenario where the end-of-period private debt (which will also be the total public debt, since the EU public debt will have been repaid in 35 years) remains as it is today, namely, $b_t^p \equiv b_{t+T} \equiv 1.71$. Then, equation (B.3) implies $s_t^p \cong 0.^{13}$ In other words, the total primary fiscal surplus required for sustainability if the public debt after 35 years simply remains as it is today (namely, 171% of GDP) is 2.3% (2.3 + 0 = 2.3), which should be compared to 1.7% in the experiment above which did not take into account the obligations of Greece to the EU institutions. If, on other hand, again as we did above, the public debt after 35 years is assumed to be 100% of GDP, the same calculations imply $s_t^p \cong 0.017^{14}$ In other words, the total primary fiscal surplus required for sustainability in the more ambitious case in which the public debt after 35 years will be 100% of GDP, is 0.023 + 0.017 = 0.04 or 4%, which should be compared to 3.4% in the experiment above which did not take into account the obligations of Greece to the EU institutions.

In sum, the primary fiscal surplus should be systematically higher (by around 0.6 percentage points) than that implied by standard-type calculations which do not take into account the obligations of the country to the EU, specifically, that the outstanding debt to ESM, EFSF, etc, has been agreed to be paid back by around 2070. This happens because, in the standard-type calculations found in most policy reports, the surplus needs to adjust merely to the differential between the real interest rate and the economy's growth rate. By contrast, when there are debt repayment deadlines like in the case of Greece, the surplus should adjust, not only to this differential but also to the extra expenditure due to repayment of the EU debt. This is a methodological issue which means that –

$$\begin{split} ^{11} \text{Thus, } s^{eu} \text{ solves } 0 &= -s^{eu} \left(\frac{1-0.98^{35}}{1-0.98} \right) + \left(0.98^{35}x0.7x1.71 \right). \\ ^{12} \text{Thus, now } s^{eu} \text{ solves } 0.7x1.71 &= -\frac{s^{eu}}{1.01} \left(\frac{1-\left(\frac{1}{1.01}\right)^{36}}{1-\frac{1}{1.01}} \right) + 0. \\ ^{13} \text{Thus, } s^p \text{ solves } 0.3x1.71 + \frac{0.023}{1.01} \left(\frac{1-\left(\frac{1}{1.01}\right)^{36}}{1-\frac{1}{1.01}} \right) = \frac{s^p}{1.01} \left(\frac{1-\left(\frac{1}{1.01}\right)^{36}}{1-\frac{1}{1.01}} \right) + 1.71 \left(\frac{1}{1.01}\right)^{36}. \\ ^{14} \text{Thus, now } s^p \text{ solves } 0.3x1.71 + \frac{0.023}{1.01} \left(\frac{1-\left(\frac{1}{1.01}\right)^{36}}{1-\frac{1}{1.01}} \right) = \frac{s^p}{1.01} \left(\frac{1-\left(\frac{1}{1.01}\right)^{36}}{1-\frac{1}{1.01}} \right) + \left(\frac{1}{1.01}\right)^{36}. \end{split}$$

irrespectively of the primary fiscal surplus implied by the standard calculations that unavoidably depend of the assumed values of the real interest rate and the economy's growth rate – the inclusion of these repayments increases the required surplus by around 0.5 percentage point on average over time. Note that all this happens even if the EU debt is paid back at policy interest rates which are lower than the market ones. Actually, our calculations have shown that had the interest rate on the EU debt been the same as the market one, the required primary fiscal surplus would have been considerably higher (around 1.7 percentage points higher on average over time. other things equal).

4 The interest rate-growth rate differential: a look at the data

As shown above, assumptions about the interest rate-growth rate differential rate are crucial for stability and hence the fiscal primary balance needed for sustainability. But what happens in practice? Table 1 reports data for the real interest rate on 10-rear sovereign bonds, the real growth rate and their resulting difference (the so-called r - g differential) in 18 EA countries. These are averages of annual data over 2001-2022 for each country. The interest rates are those in the secondary market. The numbers in parentheses for Cyprus, Greece, Ireland and Portugal exclude the sovereign debt crisis years during which these countries were shut down from sovereign bond markets and had to resort to official financial aid from the EC, the ECB and the IMF.¹⁵

As can be seen in the third column, which covers the full euro period, growth rates have exceeded interest rates in most countries except in Greece, Italy and Portugal, where the differential has been unfavourable. However, once we exclude the sovereign debt crisis years as defined above, the differential ceases to be positive in Greece and Portugal and becomes even more negative in Cyprus and Ireland (see the numbers in parentheses in the third column). Thus, at first sight, things are not bad, with the exception of Italy where the differential has been clearly unpleasant. However, the last column repeats the same exercise except that now we cover the period 2001-2014 only, namely we leave aside the period of the ECB's large-scale purchases of sovereign bonds (the so-called quantitative easing, QE) that started officially in the beginning of 2015 as well as the recent year of 2022 during which high inflation has led to negative real interest rates in most countries. Comparison of the numbers in the last two columns reveals that, in most cases, the interest rate-growth rate differential turns from negative to positive, or to less negative, in the last column, which illustrates the beneficial effect of the ECB's massive bonds purchases on bonds prices and their yields. Since such large-scale QE policies cannot continue for ever, now things look worse.

 $^{^{15}}$ That is, for these countries, we have recalculated the averages excluding their debt crisis years during which their nominal interest rate on sovereign bonds in the secondary market exceeded 6 percent. These years are 2012-14 for Cyprus, 2010-17 for Greece, 2011-12 for Ireland and 2011-13 for Portugal.

Therefore, the evidence is mixed with both positive and negative differentials over time and across countries. Also, if we think of the period since 2015 as being temporary, in the sense that sooner or later the ECB will embark on a gradual quantitative tightening, and that high inflation as a result of Russia's invasion of Ukraine will not continue to erode the real interest rates, then unfavourable differentials can be expected in several counties especially if economic growth slows down.

			Interest Ra	te – Growth
Country	Real Interest	Real Growth	Differential	
	Rate	Rate	2001-2022	2001-2014
Austria	0.3	1.5	-1.2	0.1
Belgium	0.4	1.6	-1.2	0.2
Cyprus	2.5(1.9)	2.6(3.6)	-0.1(-1.7)	1.7(-0.7)
Finland	0.7	1.4	-0.6	0.5
France	0.9	1.2	-0.3	0.8
Germany	0.4	1.2	-0.7	0.6
Greece	4.4(1.3)	0.4(2.3)	4.0(-0.4)	5.1(-0.3)
Ireland	1.4(1.0)	5.5(6.0)	-4.1(-5.0)	-0.2(-1.1)
Italy	1.6	0.3	1.3	2.4
Latvia	-0.2	3.4	-3.6	-2.5
Lithuania	0.4	4.0	-3.6	-1.7
Luxembourg	0.1	2.6	-2.5	-1.7
Malta	1.2	4.0	-2.7	-0.7
Netherlands	0.1	1.5	-1.4	0.3
Portugal	2.1(1.3)	0.8(1.3)	1.2(0)	3.0(1.4)
Slovakia	-0.3	3.5	-3.7	-3.3
Slovenia	0.8	2.5	-1.6	0.2
Spain	1.0	1.4	-0.5	0.6

Table 1

Source: Eurostat, OECD and World Government Indicators.

5 The EC's fiscal sustainability indicators

Before we move on, it is useful to clarify how standard debt arithmetic like the above relates to the EC's S1 and S2 indicators usually used for the evaluation of long-term fiscal sustainability in EU countries (see e.g. European Commision (2023, chapter 3)).

We start with S2 since, according to the EC, "the S2 indicator is the central element of the long-term sustainability analysis" (see European Commission (2022a, p. 97)). Recall first that working with this indicator presupposes R > 1. Also recall that our equations (5)-(6) above are like equations (7)-(9) in European Commission (2023, Annex A5.4). More specifically, if we rewrite our

equation (6) as $s = \left(\frac{R\left(1-\frac{1}{R}\right)}{1-\left(\frac{1}{R}\right)^{T+1}}\right) \left(b_{t-1} - \left(\frac{1}{R}\right)^{T+1} b_{t+T}\right)$, this is like equation

(11) in that Annex (except that here we include the debt ratio at the end of the projection period and we also leave aside ageing costs, etc, which enter separately the EC's formula). Notice that our s in equation (6) corresponds to the sum of the structural primary balance at the departure point (denoted as PB_{t_0} in the EC's Annex) and the additional fiscal adjustment that would be required for debt sustainability (which is the EC's S2 indicator). In other words, one should be careful how to read country rankings based on S2. The value of the latter depends crucially on the initial budgetary position, PB_{t_0} , which, in the EC's calculations, is assumed to remain constant and unchanged during all years of the experiment. But this means that if PB_{t_0} is say 1.8%, which is the forecasted value for the 2023 structural primary balance in Greece as provided by the EC's (2023) forecasts, then S2 < 0 since $s \approx 0.017 = 0.018 + S2$,¹⁶ while, if PB_{t_0} is say -1.7%, which is the forecasted value for Germany or the Euro Area as a whole for 2023, then $S_2 > 0$, since $s \approx 0.017 = -0.017 + S_2$. That is, the initial budgetary position is critical to the extent that it is assumed to remain unchanged throughout the calculations. In other words, if a country happens to enjoy a structural primary surplus in the departure year, this naturally means that the extra fiscal effort can be small or even negative in the years to come; and vice versa.¹⁷ This methodological issue (jointly with forecasted developments in ageing costs, etc) can perhaps explain why Greece is ranked as a low fiscal risk country in the long run, while, Germany is ranked as a medium fiscal risk country and Luxembourg as a high fiscal risk country in the EC's (2023, chapter 3) Debt Sustainability Monitor report. This ranking is rather counter-factual.

The same concerns apply to the S1 indicator. Recall first that working with this indicator presupposes R < 1 which means that we work with equation (4). Also recall that our equation (3)-(4) above are like equations (4)-(5) in European Commision (2023, Annex A5.3). In particular, if we rewrite our equation (4) as $-d = s = \left(\frac{R^t(1-R)}{1-R^t}\right)b_0 - \left(\frac{1-R}{1-R^t}\right)b_t$, this is like equation (6) in that Annex (except that again here we leave aside ageing costs, etc, which enter separately the EC's formula). Then, the same remark as in the case of S2 applies. Namely, one should be careful how to read S1 since its value depends crucially on the initial budgetary position, PB_{t_0} , which, in the EC's calculations, is assumed to remain constant and unchanged during all years of the experiment. Also, for the same reasons as in the case of S2, this is why, if we use S1 as a criterion for fiscal sustainability, Greece is again ranked as a low fiscal risk country in the long run, while, Germany is ranked as a medium fiscal risk country and Luxembourg as a high fiscal risk country by the EC.

¹⁶Recall that $s \approx 0.017$ was our baseline solution above.

 $^{^{17}}$ The European Fiscal Board has also pointed out that "the EC's estimates are subject to notorious uncertainty surrounding output gap estimates" (see European Fiscal Board (2022, chapter 2, p. 36)).

6 The connection between debt-based fiscal rules and debt arithmetic

As said in subsection 2.3 above, in the case in which the path of public debt is unstable, there are two ways to restore stability. First, to solve the debt equation (2) forward and work with the IGBC as we did above and, second, to allow for fiscal reaction to inherited public debt. Here, we study the latter. In particular, we clarify the connection between debt arithmetic and debt-based fiscal rules, and how this connection affects the debt dynamics.

Following D'Erasmo et al (2016), let us say that the primary fiscal deficit, d_t , is contingent on outstanding public debt, that is, $d_t \equiv d_0 - \mu b_{t-1}$, where the term d_0 includes determinants of the primary balance such as exogenous factors and counter-cyclical fiscal policies, while μb_{t-1} is the debt-contingent part of the policy instrument with $\mu \geq 0$ being a feedback policy coefficient (thus, this is a Taylor-type fiscal rule). Say that the interest rate exceeds the growth rate so that the coefficient on outstanding debt is higher than 1 in equation (2) above, $R_t > 1$; then, if μ is set high enough so as $(R_t - \mu) < 1$, equation (2) can become stable from unstable meaning that the debt arithmetic changes from unpleasant to pleasant. Note however that, even in the favorable case in which the growth rate exceeds the interest rate in the first place, $R_t < 1$, a feedback reaction to outstanding debt can help the economy to converge to a lower public debt ratio and at a faster pace other things equal.¹⁸

But, of course, feedback reactions to debt imbalances are not a free lunch. A fiscal reaction like $-\mu b_{t-1}$ represents a fiscal cost, namely, the deficit has to become smaller or the surplus has to become larger other things equal. And this cost needs to be compared to the benefit from the switch to a more favorable debt arithmetic.

Formally, if $R_t = (R_t - \mu) < 1$, while $R_t > 1$, we can use equation (4) insted of (6) and solve for d_0 but, on the other hand, now the actual primary fiscal deficit needed to support this policy is $d_t \equiv d_0 - \mu b_{t-1}$ instead of d_0 only. In other words, now d_0 solves (we again assume that \tilde{R}_t is constant over time):

$$b_t \equiv d_0 \left(\frac{1 - \widetilde{R}^t}{1 - \widetilde{R}} \right) + \widetilde{R}^t b_0 \tag{8}$$

and in turn the path of d_t follows from:

$$d_t \equiv d_0 - \mu b_{t-1} \tag{9a}$$

which, using the government budget constraint, $b_t \equiv Rb_{t-1} + d_0$, implies by repeated backward substitutions that, at some time T, the primary fiscal deficit will be:

¹⁸Thus, from equation (2), in the long run, we have $(1 - R + \mu)b = d_0$ or $b = \frac{d_0}{(1 - R + \mu)}$. so the higher is μ , the lower b can be other things equal.

$$d_T = \left[1 - \mu - \mu \sum_{i=1}^{T-2} \widetilde{R}^i\right] d_0 - \mu \widetilde{R}^{T-1} b_0 =$$
$$= \left[1 - \mu - \frac{\mu \widetilde{R} \left(1 - \widetilde{R}^{T-2}\right)}{1 - \widetilde{R}}\right] d_0 - \mu \widetilde{R}^{T-1} b_0 \tag{9b}$$

where b_0 is the initial debt to GDP ratio.

What does all this imply numerically? Say that initially we had R = 1.01 > 0as in the unpleasant case studied in subsection 4.1 above, but now, thanks to the feeback reaction to debt with $\mu = 0.02$, we switch to the pleasant case, $R_t \equiv (R_t - \mu) = 0.99 < 1$. Let us also repeat the same policy experiment as above where the public debt is initially 171% and we want to reduce it to 100% in 35 years from now. Then, as we have already seen in subsection 4.1, and as also follows from (8), we have $s_0 \equiv -d_0 \approx 0.007$. Then, using these values, equation (9b) can give the time path of primary deficits. For example, in ten periods from now, we have $s_{10} \equiv -d_{10} = 0.037$, in twenty periods $s_{20} \equiv -d_{20} = 0.033$, in 30 periods $s_{30} \equiv -d_{30} = 0.029$, etc. These numbers should be compared to $s = -d \approx 0.034$, which is the primary fiscal surplus as share of GDP required when debt sustainability is achieved by a constant or flat over time, non-debt contingent policy when R is 1.01 (see subsection 4.1 above). In other words, there is an intertemporal tradeoff as typically happens in cases of reforms. If the government follows the same, flat fiscal policy in each period, it spreads its fiscal cost out equally over time or across generations.¹⁹ If, on the other hand, it follows a debt-contingent fiscal policy, it front-loads the cost of the fiscal adjustment, with higher short-term costs in terms of surpluses and smaller sacrifices in the later periods (this happens because of the more favorable interest rate-growth differential).

Note that similar results follow when R = 0.99 < 1 so that we start with the pleasant case. That is, now, again with $\mu = 0.02$, the differential becomes even more favorable, $\tilde{R}_t \equiv (R_t - \mu) = 0.97$. Then, working similarly, we get $d_0 \cong 0.019$ from (4) and in turn, using (9b), we have $s_{10} \equiv -d_{10} = 0.01$ in ten years from now, $s_{20} \equiv -d_{20} = 0.005$ in twenty years, $s_{30} \equiv -d_{30} = 0.0024$ in thirty years, etc. These numbers should be compared to $s = -d \cong 0.007$, which is the primary fiscal surplus as share of GDP required when debt sustainability is achieved by a constant over time, non-debt contingent policy (see subsection 4.1 above)

In sum, a debt-contingent fiscal policy - according to which fiscal intruments react to public debt imbalances at a constant rate - front-loads the costs of fiscal adjustment but, after a point in time, the required surpluses get smaller and smaller as the benefits of a more favorable interest rate-growth differential build up. More loosely speaking, the main benefit from front-loading is credibility of fiscal policy and hence lower interest rates. The main risk of front-loading is that fiscal austerity may lead to a recession and vicious cycles in the short

¹⁹Perhaps there is an analogy between this and Barro's (1979) tax smoothing result.

term (see CESifo (2014, chapter 3), for a richer discussion of the intertemporal tradeoffs of fiscal adjustments).

7 Is public debt arithmetic reliable?

Calculations like the above, based on the government budget constraint only, are popular in policy papers but are sensitive to assumptions about sovereign interest rates and growth rates over time. More importantly, sovereign interest rates and growth rates are endogenous variables, and the same applies to several items included in the primary fiscal balance (think of tax revenues, social expenditure programs, etc). In reality, all these variables are endogenous and hence - in the absence of Ricardian Equivalence - depend, directly or indirectly, on the inherited public debt itself. Such endogeneity implies that the debt dynamics, and hence what is needed for debt stability and fiscal sustainability, are more complicated than those implied by the above popular policy analysis. Note that the same applies when we introduce fiscal reaction functions to restore stability as we did above; the behaviour of economic agents is affected by fiscal (re)actions and this can again shape the growth rate, the interest rate, tax bases, etc. All this, as pointed out by D' Erasmo et al (2016), is a reflection of the Lucas critique.²⁰

The above imply that a reliable quantitative fiscal sustainability analysis requires a structural approach (see also e.g. Alesina and Passalacqua (2016) and D' Erasmo et al (2016)). This means the use of macroeconomic models where these three key drivers of public debt dynamics (the sovereign real interest rate, the growth rate of real GDP and most items included in the primary fiscal balance) are all endogenous variables whose paths over time are affected - among other things - by the accumulated public debt as well as by policy reactions (if any) to it. In other words, in such models, private decisions, and in turn macroeconomic outcomes like growth rates and market interest rates, are not invariant with respect to state variables and policy actions. There are many quantitative dynamic general (dis)equilibrium models of this type in the academic literature but also by researchers in the EC, the ECB, the IMF, etc. The QUEST model used by the EC over the years is a well-known example.

Then, to the best of our understanding, there are some common messages from the macroeconomic literature: First, given the current situation, in almost all cases, if a shock hits the economy, macroeconomic stability and determinacy can be guaranteed only if some fiscal policy instruments react systematically to public debt imbalances and, specifically, if these instruments react to deviations of the outstanding public debt to GDP ratio from a policy target value.²¹ In

 $^{^{20}}$ See e.g. Sargent (2023) for a recent technical paper on the Lucas critique.

²¹For the US economy, see e.g. Leeper et al (2010), Davig et al (2010), Davig and Leeper (2011) and Malley and Philippopoulos (2023). For the Euro Area as a whole, see e.g. Dimakopoulou et al (2023) and the references therein, while see e.g. Malley et al (2009) for a study of the big EU countries before the global financial crisis. For the Greek economy, see e.g. Papageorgiou (2014), Dellas et al (2017), Economides et al (2021, 2022), Dimakopoulou et al (2022) and, for an econometric model, Dendramis et al (2022).

other words, as Sims (2017) points out, to ensure stability and hence get a solution, we cannot assume that government spending and/or tax rates are independent of the public debt path.²² Second, it is hard to find self-financing fiscal expansions even when the latter are in the form of an increase in public investment. In other words, even when an increase in public debt is used to finance an increase in public investment, which augments public infrastructure and enhances economic growth and tax bases in the medium term, a cut in another public spending item, and/or a rise in a tax rate, are also unavoidable at least in the early period during which the public debt is rising.²³ Third, the effect of a cut in public spending, or a rise in tax rates, on the debt to GDP ratio is far from obvious. Depending on the instrument used for fiscal consolidation, the latter may reduce the level of debt but it can also reduce the level of GDP so that the debt ratio can increase (for recent empirical evidence and how to tackle soaring public debt, see International Monetary Fund (2023, chapter 3)). In other words, one should carefully check the size of the multiplier of each policy instrument used to bring the debt ratio down.²⁴

But, regarding the first message, what happens in practice? Do we observe fiscal reactions to rising public debt? This is addressed next.

8 Fiscal reaction functions: a look at the data

Although it is widely recognized that debt-contingent fiscal policy rules are needed for stability in structural macroeconomic models, the empirical evidence is mixed to say the least. The European Commission itself reports the lack of fiscal reaction to public debt imbalances (see European Commission (2015, part IV; and 2021, part IV)). Actually, as it admits, "the debt ratio in particular does not seem to have played any role in determining the fiscal effort required, which is interesting considering not only that debt is the centre of focus of the existing literature on the fiscal reaction function but also considering the legislation itself, which states that the medium-term debt position (its dynamics and sustainability) constitutes the key factor in determining the recommendation" (see European Commission (2021, p. 132). D' Erasmo et al (2016) also provide estimates of fiscal reaction functions for a number of countries and find that debt stabilization reactions become much weaker when post-2008 are added to the sample. In particular, they report that there is a structural break after 2008 in the response of the primary balance to high debt both in the European economies and the USA. On the other hand, Attinasi et al (2019) provide evi-

 $^{^{22}}$ Sims (2017) uses an educational simple general equilibrium model to show how feedback fiscal rules (in the sense that tax rates rise, and/or public spending falls, in response to rising public debt) can avoid an explosive path of debt. As he elaborates, this response should be high enough so that the debt ratio does not explode upward but, at the same time, not too high that the debt ratio explodes downward (see also our third message below).

 $^{^{23}{\}rm See}$ e.g. Malley and Philippopoulos (2022) and the references there for the US economy. See e.g. Dimakopoulou et al (2022) for the Greek economy.

 $^{^{24}}$ See e.g. Philippopoulos et al (2017) for various fiscal policy scenaria used to bring the public debt ratio down. In the final section, we summarize some related results.

dence that, in high-debt EA countries, cyclically-adjusted primary balances do react to inherited public debt, although this is significant only during bad times when the output gap is negative (see their Table 7 in particular).

We also provide are own evidence. In Table 2, we calculate the correlation between current public debt as share of GDP and next year's primary fiscal surplus as share of GDP in 18 EA countries.

Table 2					
Correlation between public debt and next year's primary fiscal surplus					
	Public debt to GDP	Correlation			
Country	(average 2001-2022)	$(\ 2001-2022)$			
Austria	75.5	-0.26			
Belgium	101.7	-0.13			
Cyprus	78.7	0.22 (*)			
Finland	55.0	-0.66			
France	85.2	-0.44			
Germany	68.7	0.39			
Greece	149.4	0.19 (*)			
Ireland	60.8	-0.07			
Italy	123.7	-0.55			
Latvia	30.7	-0.01			
Lithuania	31.3	0.30			
Luxembourg	16.7	-0.11			
Malta	60.3	0.20			
Netherlands	55.2	-0.32			
Portugal	101.8	0.39(*)			
Slovakia	46.2	0.31			
Slovenia	51.0	0.08			
Spain	77.0	-0.21			

Source: Eurostat and own calculations.

As can be seen in the last column of Table 2, and for most of the EA countries, the correlation is negative meaning that an increase in public debt to GDP ratio in the current period is associated with a lower primary fiscal surplus, or a higher primary fiscal deficit, in the next period. Exceptions include Germany, which has a relatively high positive coefficient, as well as Cyprus, Greece and Portugal (these three countries are marked with an asterisk). However, recall that in the 2010s Cyprus, Greece and Portugal had been in enforced fiscal austerity programs as a condition for their official bailout from EU public institutions. We additionally report that, for the EA as a whole, the correlation coefficient is also negative, around -0.2.

In sum, we think it is fair to say that there is little evidence of systematic stabilizing fiscal reaction to debt imbalances. Given this, if, in practice, we do not observe any systematic fiscal reaction to public debt imbalances, then, quoting Leeper et al (2010) in their study for the US, a natural question to ask ourselves is "Why do forward-looking agents continue to purchase bonds with relatively low interest rates?". The answer given by Leeper and his co-authors is that - to the extent that we want to maintain the assumption of rationality - agents believe that current inaction is temporary and it will be replaced by necessary policy corrections in the future. This is why trust, expectations about the future, and what is signaled by policymakers in the present, are crucial.

9 OK, fiscal rules, but which ones?

As said above, a common property of most structural macroeconomic models is that Taylor-type feedback rules, according to which tax-spending instruments respond to outstanding public debt among other state variables, are needed to avoid an explosive path of public debt. As also said above, this is the type of fiscal rules employed in most research papers. Other types of fiscal rules usually met in policy circles, which are believed to limit the accumulation of excessive deficits and debts, include numerical targets (like a balanced budget rule, a debt ceiling, a limit on public spending, etc) or the so-called golden rule according to which budget deficits are allowed to finance public investment only (see e.g. Alesina and Passalacqua (2016) for a review of fiscal rules).

The debate for fiscal rules has been particularly hot in the EU. Since the Maastricht Treaty in 1992, the agreement has been that, in a second-best world, fiscal rules at national level are needed for the viability of the single currency. Various rules have been introduced and debated over the years without much success or agreement.²⁵ It is fair to say that the EU's fiscal rules, past and present, have been of four kinds: deficit-based, debt-based, expenditure-based and structural balance-based (see e.g. European Commission (2021, part IV)). Most of the time so far, the policy emphasis and compliance have been on deficit-based rules like the 3% ceiling. By contrast, in the new economic governance framework recently presented by the European Commission (2022b and 2023b), although references to the 3% ceiling for fiscal deficits and the 60% target for the debt ratio remain, the empasis has shifted from limits on budget deficits to limits on public expenditure.

To the best of our undestanding, the recently proposed expenditure rule by the EC works as follows. We first calculate the country's annual growth rate of primary government expenditure, net of discretionary revenue measures and one-offs. Second, we calculate the ten-year average growth rate of potential nominal GDP minus the convergence margin necessary to ensure an adjustment of the country's structural budget deficit of the general government in line with the country's structural balance rule (this adjustment implies that we take into account whether the structural budget balance of the general government is above, or below, the country's Medium-Term Objective, MTO). Finally, if the country's annual growth rate of primary government expenditure from the first

²⁵ For the history of EU fiscal rules as well as for the current state of affairs and controversies, see e.g. Wyplosz (2014, 2021, 2023), Beetsma and Larch (2019), Bilbiie et al (2021), Beetsma (2022) and the references cited there.

step exceeds the adjusted ten-year average growth rate from the second step, a fiscal correction or effort is needed. Before we assess this rule, we should recall, as Wyplosz (2023) points out, that the expenditure rule is not new. It was introduced in 2011 as part of the preventive arm of the SGP; the difference is that now the EC proposes to make it the only, or the main, measure to be used.

The general idea behind the emphasis on an expenditure rule is that primary public expenditures are directly under the control of national governments so such a rule has to do with a policy instrument rather than an intermediate target like the budget balance or the cyclically adjusted budget balance which have been the main measures in the past. As such, the rule has been received positively (see e.g the European Central Bank (2023) and Wyplosz (2023)). On the other hand, it has also received criticism. For example, the suggested expenditure rule is too complex; it has to rely on unobservable variables that are not readily available (like cyclical unemployment, potential growth, discretionary revenue changes, etc); its methodology needs clarification; etc (again see e.g the European Central Bank (2023) and Wyplosz (2023)). In other words, the usual broader concerns about simplicity, transparency and effectiveness of policy rules continue to apply.

The discussion about the optimality, or simply the effectiveness, of the EU's fiscal rules (previous and new) has been big and we cannot review it here (see the references above for good reviews). Nevertheless, to this literature, we wish to add three points that might be useful.

Our first point is obvious and general but we feel that sometimes is forgotten in the heat of political debates. All policy rules used in practice are, by definition, suboptimal (and this includes the famous Taylor rule for monetary policy). Ramsey, and especially time-consistent, optimal policy rules would be too complicated, even computationally, to be useful for policy-making since they follow from a solution of a large general equilibrium model where the government acts as a Stackelberg leader which means that a complete description of optimal policy results in complicated feedback rules where policy instruments react to a very large number of state variables including auxilliary multipliers (see e.g. Benigno and Woodford (2006) and Schmitt-Grohe and Uribe (2006) among many others). This is why policymakers need to resort to "simple and implementable" rules meaning that policy instruments react to a small number of observable variables only (see e.g. Schmitt-Grohe and Uribe (2006, 2007) and Kliem and Kriwoluzky (2014)). On the other hand, although the EU's fiscal rules cannot, by definition, be immune to the standard criticism about optimality, the calculation of the recommended maximum growth rate of primary public expenditure relies on too many unobservable variables so it is hard to be characterized as "simple and implementable".

Our second point is that, as already reported above, the lesson from most structural macroeconomic models is that macroeconomic stability requires policy reaction to the public debt ratio itself. We report that, by making use of rather conventional DSGE models like those listed in section 7 above, our simulation experiments systematically imply that by simply keeping the deficit below the numerical value of 3%, or by restricting the growth rate of primary expenditure, or by reacting to flow variables (like the budget balance, or the cyclically adjusted balance, or the MTO) do not seem able to restore stability in an otherwise unstable economy. And this is hardly surprising: public debt is a state stock variable so, if it happens to be explosive, reaction to this very variable is necessary to restore dynamic stability. This means that a policy rule should be contingent (perhaps among other things as discussed next) on the outstanding public debt ratio or the deviation of the latter from a target value, at least for some time. This is simply necessary for stability.

Third, a measure of fiscal stance - and it is the fiscal stance that dictates the macroeconomic variables that policy instruments should be contingent on should include those variables that shape the dynamics of the debt ratio. And, as we saw in detail in section 2 above and as is well known, these variables are the primary fiscal balance, the outstanding public debt to GDP ratio and the interest rate-growth rate differential. Among them, reaction to the outstanding debt ratio is necessary, for the reasons said just above. But reaction to the other two (which could make the fiscal effort smaller, or bigger, depending on their evolution) can be desirable in terms of the required fiscal adjustment. For example, if the interest-rate growth rate differential is favorable so that the economy grows out its debt ratio, fiscal reaction to public debt (which would mean unpopular, and perhaps recessionary, fiscal consolidation) can be milder. In other words, the fiscal stance could be a weighted average of the gap between the debt ratio and its target value, the gap between the primary fiscal balance and its target value, and the difference between the sovereign interest rate and the economy's growth rate. In turn, the expenditure instrument(s) should be contingent on this stance.

Formally, the fiscal stance can be defined as:²⁶

$$\left(\frac{b_{t-1}}{b_t^{t\,\arg et}}\right)^{a_1} \left(\frac{d_t}{d_t^{t\,\arg et}}\right)^{a_2} \left(\frac{r_t}{\gamma_t}\right)^{1-a_1-a_2} \le 1 \tag{10}$$

where $0 < a_1, a_2 < 1$ are policy weights that the EU can decide on, while the policy targets, again decided by the EU, could be, for example, $b_t^{t \arg et} =$ $0.95b_{t-1}$ in case we want the debt ratio to fall over time and $d_t^{t \arg et} = MTO$. In turn, public spending (being a component of d_t) can follow residually to satisfy the above, written as a binding equality, to give a relatively simple and implementable feeback fiscal rule for primary expenditure.

10 Policy conclusions

Sovereign interest rates, growth rates and most items included in primary fiscal balances are all endogenous variables which are jointly determined. It is also obvious that all of them are affected by economic policies. This is another application of the Lucas critique, which naturally implies that approaching the

²⁶Recall that $r_t = i_t - \pi_t$ is the real interest rate on sovereign bonds and γ_t is the growth rate of real GDP. It is important to report that that this is very similar to Korea's recent fiscal rules.

issue of fiscal sustainability through the lens of the government budget constraint only is not reliable for quantitative conclusions.

Therefore, a more reliable analysis of fiscal sustainability necessitates the use of structural dynamic general (dis)equilibrium macroeconomic models. In these models, all the above key variables that shape the public debt dynamics are endogenous variables and, as such, are affected by policy actions and the public debt itself. Then, a common finding from this literature is that if we assume that fiscal policies remain unchanged as in the current data or more generally are exogenous, the path of public debt is explosive over time and this applies to most countries. Hence, debt-contingent fiscal rules are necessary according to which fiscal instruments react to the gap between the outstanding public debt and a policy target value. We also contributed to the debate on fiscal rules in the EU by suggesting that, in addition to the public debt gap which is necessary for stability and fiscal sustainability, fiscal rules should also be contingent on the interest rate-growth rate differential and the gap between the primary fiscal balance from its medium-term objective. Such a rule can give more flexibility and explicitly reward those countries that, although have high public debt, they manage to grow it out and/or enjoy the trust of the markets as reflected in relatively low sovereign interest rates.

Finally, we should recall that which particular fiscal policy instrument is being used to bring public debt down is essentially a fiscal policy multiplier problem. The macroeconomic literature (see e.g. Philippopoulos et al (2017) and the references therein) suggests that a damage-minimizing policy mix is the one in which we use fiscal instruments with small output multipliers to bring public debt down and - once public debt has been brought down - we allow fiscal instruments with large output multipliers to take advantage of the fiscal space created; the anticipation of the latter, if credible, shapes private incentives and may mitigate the recessionary effects even in the short term. This is consistent with the "expenditures" rules suggested recently by the EC, although one has to be clear regarding the kind of public expenditures that should be used to stabilize public debt.

References

- Alesina A. and A. Passalacqua (2016): The political economy of government debt, in Handbook of Macroeconomics, volume 2B, edited by J.. Taylor and H. Uhlig, North Holland.
- [2] Alesina A., C. Favero and F. Giavazzi (2019): Austerity, Princeton University Press.
- [3] Attinasi M. G., A. A. Palazzo and B. Pierluigi (2019): Fiscal activism in the EA and in other advanced economies: new evidence, European Central Bank, Working paper no. 2344.
- [4] Azariadis C. (1993): Intertemporal Macroeconomics, Blackwell.

- [5] Barro R. (1979): On the determination of the public debt, Journal of Political Economy, 87, 940-971.
- [6] Bassetto M. and D. Miller (2022): A monetary-fiscal theory of sudden inflations, Staff Report no. 641, Federal Reserve Bank of Minneapolis.
- [7] Beetsma R. and M. Larsh (2019): EU fiscal rules: Further reform or better implementation?, ifo DICE Report, 17, 7-11.
- [8] Beetsma R. (2022): The economics of fiscal rules and debt sustainability, Intereconomics, 57, 11-15.
- Benigno P. and M. Woodford (2006): Optimal taxation in an RBC model: A linear-quadratic approach, Journal of Economic Dynamics and Control, 30, 1445-1489.
- [10] Bilbiie F., T. Monacelli and R. Perotti (2021): Fiscal policy in Europe: Controversies over rules, mutual insurance and centralization, Journal of Economic Perspectives, 35, 77-110.
- [11] Blanchard O. and S. Fischer (1989): Lectures on Macroeconomics, MIT Press.
- [12] Blanchard O., C. Chouraqui, R. Hagenmann and N. Sartor (1990): The sustainability of fiscal policies: New answers to an old question, OECD Economic Studies, 15, 7-36.
- [13] Blanchard O. and M. Das (2017): A new index of debt sustainability, NBER, Working Paper no. 24068.
- [14] Blanchard O. (2019): Public debt and low interest rates, American Economic Review, 109, 1197-1229.
- [15] Buiter W. and K. Kletzer (1992): Who's afraid of the public debt? American Economic Review, 82, 290-298.
- [16] Buiter W. (2021): Central Banks as Fiscal Players, Cambridge University Press.
- [17] CESifo (2014): The EEAG Report on the European Economy, CESifo.
- [18] Davig T., E. Leeper and T. Walker (2010): Unfunded liabilities and uncertain fiscal financing, Journal of Monetary Economics, 57, 600-619.
- [19] Davig T. and E. Leeper (2011): Monetary-fiscal policy interactions and fiscal stimulus, European Economic Review, 55, 211-227.
- [20] Dellas H., D. Malliaropulos, D. Papageorgiou and E. Vourvachaki (2017): Fiscal policy with an informal sector, Bank of Greece, Working Paper, no. 235.

- [21] Dendramis Y., G. Dimitrakopoulos and E. Tzavalis (2022): Assessing downside public debt risks in an environment of negative interest rates growth differentials: A structural policy scenario analysis, mimeo, Athens University of Economics and Business.
- [22] D' Erasmo P., E. Mendoza and J. Zhang (2016): What is sustainable public debt?, in Handbook of Macroeconomics, volume 2B, edited by J.. Taylor and H. Uhlig, North Holland.
- [23] Dimakopoulou V., G. Economides and A. Philippopoulos (2022): The ECB's policy, the Recovery Fund and the importance of trust and fiscal corrections: The case of Greece, Economic Modelling, 112.
- [24] Dimakopoulou V., G. Economides, A. Philippopoulos and V. Vassilatos (2023): Can central banks do the unpleasant job that governments should do?, CESifo Working Paper, no. 10603.
- [25] Dornbusch R. and M. Draghi (1990): Public Debt Management: Theory and Evidence, Cambridge University Press and CEPR.
- [26] Economides G., D. Papageorgiou and A. Philippopoulos (2021): Austerity, assistance and institutions: Lessons from the Greek sovereign debt crisis, Open Economies Review, 32, 435-478.
- [27] Economides G., A. Philippopoulos and V. Vassilatos (2022): The impact of the lockdown on the Greek economy and the role of the Recovery Fund, The B.E. Journal of Macroeconomics, 22, 307-361.
- [28] Economides G. and A. Philippopoulos (2023): Fiscal sustainability: Interest rates, growth and debt-based policy rules, EconPol Forum, CESifo, 24, 11-15.
- [29] Eichengreen B. (2022): The debate over US public debt: Overburdened or overwrought?, CESifo Forum, 23, 1, 3-8.
- [30] European Central Bank (2023): Opinion of the ECB on a proposal for economic governance reform in the Union, 5.7.2023.
- [31] European Commission (2015):Public Finances in EMU.
- [32] European Commission (2021): Public Finances in EMU.
- [33] European Commission (2022a): Fiscal Sustainability Report 2021, Institutional Paper no. 171, April 2022.
- [34] European Commission (2022b): Post-programme Surveillance Report, Greece Autumn 2022, Institutional Paper no. 191, November 2022.
- [35] European Commission (2022c): Economic Forecasts, Autumn 2022.

- [36] European Commission (2022d): A communication on orientations for a reform of the EU economic governance framework, November 2022.
- [37] European Commission (2023a): Debt Sustainability Monitor 2022, Institutional Paper no. 199, April 2023.
- [38] European Commission (2023b): Fiscal statistical tables providing background data relevant for the accessment of the 2023 Stability or Convergence Programmes, Commission Staff Working Document, 24.5.2023.
- [39] European Fiscal Board (2022): Annual Report 2022.
- [40] Friedman B. (1992): Learning from the Reagan deficits, American Economic Review, 82, 299-304.
- [41] International Monetary Fund (2023): World Economic Outlook: A Rocky Revovery.
- [42] Jones C. (2008): Macroeconomics, Norton.
- [43] Kliem M. and A. Kriwoluzky (2014): Toward a Taylor rule for fiscal policy, Review of Economic Dynamics, 17, 294-302.
- [44] Leeper E., M. Plante and N. Traum (2010): Dynamics of fiscal financing in the U.S., Journal of Econometrics, 156, 304-321.
- [45] Malley J., A. Philippopoulos and U. Woitek (2009): To react or not? Fiscal policy, volatility and welfare in the EU-3, European Economic Review, 53, 689-714.
- [46] Malley J. and A. Philippopoulos (2022): The macroeconomic effects of funding US infrastructure, forthcoming, European Economic Review.
- [47] Obstfeld M. and K. Rogoff (1996): Fondations of International Macroeconomics, MIT Press.
- [48] Papageorgiou D., A. Philippopoulos and V. Vassilatos (2010): A toolkit for the study of fiscal policy in Greece, in Essays in Economics: Applied Studies on the Greek Economy, edited by S. Balfoussias, P. Hatzipanayotou and C. Kanellopoulos, Centre of Planning and Economic Research, 50 Years Anniversary Publication, Athens.
- [49] Papageorgiou D. (2014): BoGGEM: A DSGE model for policy simulations, Bank of Greece, Working Paper, no. 182.
- [50] Philippopoulos A., P. Varthalitis and V. Vassilatos (2017): Fiscal consolidation and its cross-country effects, Journal of Economic Dynamics and Control, 83, 55-106.
- [51] Sargent T. (1987): Macroeconomic Theory, second edition, Academic Press.

- [52] Sargent T. (2023): Critique and consequence, mimeo, New York University.
- [53] Schmitt-Grohe S. and M. Uribe (2006): Optimal fiscal and monetary policy in a medium-scale macroeconomic model, NBER Macoeconomics Annual, 383-425.
- [54] Schmitt-Grohe S. and M. Uribe (2007): Optimal simple and implementable monetary and fiscal rules, Journal of Monetary Economics, 54, 1702-1725.
- [55] Schuknecht L. (2022): Public debt The EU perspective, CESifo Forum, 23, 9-18.
- [56] Sims E. (2017): Graduate Macro Theory II, Lecture notes, University of Notre Dame.
- [57] Walsh C. (2017): Monetary Theory and Policy, 4th edition, MIT Press.
- [58] Wickens M. (2008): Macroeconomic Theory, Princeton University Press.
- [59] Wyplosz C. (2014): Fiscal rules: theoretical issues and historical experiences, in Fiscal Policy after the Financial Crisis, edited by A. Alesina and F. Giavazzi, University of Chicago Press and NBER.
- [60] Wyplosz C. (2021): Toward a top-down fiscal rule, mimeo, The Graduate Institute, Geneva.
- [61] Wyplosz C. (2023): The European Commission's expenditure benchmark, CEPR Policy Insight. no. 121.

Appendices

Appendix A

Consider the first-order linear difference equation in Y_t :

$$Y_t = RY_{t-1} + a + bX_t \tag{A.1}$$

where R < 1.

By repeated backward substitutions (see e.g. Sargent (1987, chapter IX) and Azariadis (1993, chapter 2)), we get:

$$Y_t = a \sum_{i=0}^{t-1} R^i + b \sum_{i=0}^{t-1} R^i X_{t-i} + R^t Y_0$$

where Y_0 is a given initial value.

Since R < 1, this can be written as:

$$Y_t = \frac{a(1-R^t)}{(1-R)} + b \sum_{i=0}^{t-1} R^i X_{t-i} + R^t Y_0$$
(A.2)

which, if we assume for simplicity that X is constant, simplifies to:

$$Y_t = \frac{a\left(1 - R^t\right)}{(1 - R)} + \frac{bX\left(1 - R^t\right)}{(1 - R)} + R^t Y_0 \tag{A.3}$$

or:

$$Y_t = \frac{a + bX}{(1 - R)} + R^t \left(Y_0 - \frac{a + bX}{(1 - R)} \right)$$
(A.4)

This generalizes (3)-(4) in the text.

Note that the process of the driving force, X_t , can also be important for convergence. For example, say that now $X_{t+1} = \theta X_t$, where θ is a parameter. Then, equation A.2 becomes:

$$Y_t = \frac{a(1-R^t)}{(1-R)} + bX_t \sum_{i=0}^T \left(\frac{R}{\theta}\right)^i + R^t Y_0$$
(A.2a)

so that, in this more general case, stability also requires $\left|\frac{R}{\theta}\right| < 1$ which puts additional restrictions on the exogenous variable. Recall that the exogenous variable is the primary fiscal deficit in our analysis.

Appendix B

Consider the same equation in Y_t :

$$Y_t = RY_{t-1} + a + bX_t \tag{B.1}$$

except that now R > 1.

By repeated forward substitutions (see e.g. Sargent (1987, chapter IX) and Azariadis (1993, chapter 2)), we get:

$$Y_{t-1} = -\frac{a}{R} \left(1 + \frac{1}{R} + \dots + \left(\frac{1}{R}\right)^T \right) - \frac{b}{R} \sum_{i=0}^T \left(\frac{1}{R}\right)^i X_{t+i} + \left(\frac{1}{R}\right)^{T+1} Y_{t+T}$$

or

$$Y_{t-1} = -\frac{a}{R} \left(\frac{1 - \left(\frac{1}{R}\right)^{T+1}}{1 - \frac{1}{R}} \right) - \frac{b}{R} \sum_{i=0}^{T} \left(\frac{1}{R}\right)^{i} X_{t+i} + \left(\frac{1}{R}\right)^{T+1} Y_{t+T}$$
(B.2)

which, if we assume for simplicity that X is constant, simplifies to:

$$Y_{t-1} = -\frac{a}{R} \left(\frac{1 - \left(\frac{1}{R}\right)^{T+1}}{1 - \frac{1}{R}} \right) - \frac{bX}{R} \left(\frac{1 - \left(\frac{1}{R}\right)^{T+1}}{1 - \frac{1}{R}} \right) + \left(\frac{1}{R}\right)^{T+1} Y_{t+T}$$
(B.3)

or

$$Y_t = \frac{a+bX}{(1-R)} + \left(\frac{1}{R}\right)^{T+1} \left(Y_{t+T} - \frac{a+bX}{(1-R)}\right)$$
(B.4)

This generalizes (5)-(6) in the text.

Note that the process of the driving force, X_t , can also be important for convergence. For example, say that now $X_{t+1} = \theta X_t$, where θ is a parameter. Then, equation B.2 becomes:

$$Y_{t-1} = -\frac{a}{R} \left(\frac{1 - \left(\frac{1}{R}\right)^{T+1}}{1 - \frac{1}{R}} \right) - \frac{b}{R} X_t \sum_{i=0}^T \left(\frac{\theta}{R}\right)^i + \left(\frac{1}{R}\right)^{T+1} Y_{t+T}$$
(B.2a)

so that, in this more general case, stability also requires $\left|\frac{\theta}{R}\right| < 1$ which puts additional restrictions on the exogenous variable (see also e.g. Obstfeld and Rogoff (1996, p. 729)). Recall that the exogenous variable is the primary fiscal deficit in our analysis.



Department of Economics Athens University of Economics and Business

List of Recent Working Papers

<u>2021</u>

- 01-21 Historical Cycles of the Economy of Modern Greece From 1821 to the Present, George Alogoskoufis
- 02-21 Greece Before and After the Euro: Macroeconomics, Politics and the Quest for Reforms, George Alogoskoufis
- 03-21 Commodity money and the price level, George C. Bitros. Published in: *Quarterly Journal* of Austrian Economics, 2022
- 04-21 Destabilizing asymmetries in central banking: With some enlightenment from money in classical Athens, George C. Bitros. Published in: *Journal of Economic Asymmetries*, 2021
- 05-21 Exploring the Long-Term Impact of Maximum Markup Deregulation, Athanasios Dimas and Christos Genakos
- 06-21 A regularization approach for estimation and variable selection in high dimensional regression models, Y. Dendramis, L. Giraitis, G. Kapetanios
- 07-21 Tax Competition in the Presence of Environmental Spillovers, Fabio Antoniou, Panos Hatzipanayotou, Michael S. Michael, Nikos Tsakiris
- 08-21 Firm Dynamics by Age and Size Classes and the Choice of Size Measure, Stelios Giannoulakis and Plutarchos Sakellaris
- 09-21 Measuring the Systemic Importance of Banks, Georgios Moratis, Plutarchos Sakellaris
- 10-21 Firms' Financing Dynamics Around Lumpy Capacity Adjustments, Christoph Görtz, Plutarchos Sakellaris, John D. Tsoukalas
- 11-21 On the provision of excludable public goods General taxes or user prices? George Economides and Apostolis Philippopoulos
- 12-21 Asymmetries of Financial Openness in an Optimal Growth Model, George Alogoskoufis
- 13-21 Evaluating the impact of labour market reforms in Greece during 2010-2018, Georgios Gatopoulos, Alexandros Louka, Ioannis Polycarpou, Nikolaos Vettas
- 14-21 From the Athenian silver to the bitcoin standard: Private money in a state-enforced free banking model, George C. Bitros
- 15-21 Ordering Arbitrage Portfolios and Finding Arbitrage Opportunities. Stelios Arvanitis and Thierry Post
- 16-21 Inconsistency for the Gaussian QMLE in GARCH-type models with infinite variance, Stelios Arvanitis and Alexandros Louka
- 17-21 Competition and Pass-Through: Evidence from Isolated Markets, Christos Genakos and Mario Pagliero

- 18-21 Exploring Okun's Law Asymmetry: An Endogenous Threshold LSTR Approach, Dimitris Christopoulos, Peter McAdam and Elias Tzavalis
- **19-21** Limit Theory for Martingale Transforms with Heavy-Tailed Multiplicative Noise, Stelios Arvanitis and Alexandros Louka
- 20-21 Optimal taxation with positional considerations, Ourania Karakosta and Eleftherios Zacharias
- 21-21 The ECB's policy, the Recovery Fund and the importance of trust: The case of Greece, Vasiliki Dimakopoulou, George Economides and Apostolis Philippopoulos

<u>2022</u>

- 01-22 Is Ireland the most intangible intensive economy in Europe? A growth accounting perspective, Ilias Kostarakos, KieranMcQuinn and Petros Varthalitis
- 02-22 Common bank supervision and profitability convergence in the EU, Ioanna Avgeri, Yiannis Dendramis and Helen Louri
- 03-22 Missing Values in Panel Data Unit Root Tests, Yiannis Karavias, Elias Tzavalis and Haotian Zhang
- 04-22 Ordering Arbitrage Portfolios and Finding Arbitrage Opportunities, Stelios Arvanitis and Thierry Post
- 05-22 Concentration Inequalities for Kernel Density Estimators under Uniform Mixing, Stelios Arvanitis
- 06-22 Public Sector Corruption and the Valuation of Systemically Important Banks, Georgios Bertsatos, Spyros Pagratis, Plutarchos Sakellaris
- 07-22 Finance or Demand: What drives the Responses of Young and Small Firms to Financial Crises? Stelios Giannoulakis and Plutarchos Sakellaris
- 08-22 Production function estimation controlling for endogenous productivity disruptions, Plutarchos Sakellaris and Dimitris Zaverdas
- 09-22 A panel bounds testing procedure, Georgios Bertsatos, Plutarchos Sakellaris, Mike G. Tsionas
- **10-22** Social policy gone bad educationally: Unintended peer effects from transferred students, Christos Genakos and Eleni Kyrkopoulou
- **11-22** Inconsistency for the Gaussian QMLE in GARCH-type models with infinite variance, Stelios Arvanitis and Alexandros Louka
- 12-22 Time to question the wisdom of active monetary policies, George C. Bitros
- 13-22 Investors' Behavior in Cryptocurrency Market, Stelios Arvanitis, Nikolas Topaloglou and Georgios Tsomidis
- 14-22 On the asking price for selling Chelsea FC, Georgios Bertsatos and Gerassimos Sapountzoglou
- 15-22 Hysteresis, Financial Frictions and Monetary Policy, Konstantinos Giakas
- 16-22 Delay in Childbearing and the Evolution of Fertility Rates, Evangelos Dioikitopoulos and Dimitrios Varvarigos
- 17-22 Human capital threshold effects in economic development: A panel data approach with endogenous threshold, Dimitris Christopoulos, Dimitris Smyrnakis and Elias Tzavalis
- 18-22 Distributional aspects of rent seeking activities in a Real Business Cycle model, Tryfonas Christou, Apostolis Philippopoulos and Vanghelis Vassilatos

- 01-23 Real interest rate and monetary policy in the post Bretton Woods United States, George C. Bitros and Mara Vidali
- 02-23 Debt targets and fiscal consolidation in a two-country HANK model: the case of Euro Area, Xiaoshan Chen, Spyridon Lazarakis and Petros Varthalitis
- 03-23 Central bank digital currencies: Foundational issues and prospects looking forward, George C. Bitros and Anastasios G. Malliaris
- 04-23 The State and the Economy of Modern Greece. Key Drivers from 1821 to the Present, George Alogoskoufis
- 05-23 Sparse spanning portfolios and under-diversification with second-order stochastic dominance, Stelios Arvanitis, Olivier Scaillet, Nikolas Topaloglou
- 06-23 What makes for survival? Key characteristics of Greek incubated early-stage startup(per)s during the Crisis: a multivariate and machine learning approach, Ioannis Besis, Ioanna Sapfo Pepelasis and Spiros Paraskevas
- 07-23 The Twin Deficits, Monetary Instability and Debt Crises in the History of Modern Greece, George Alogoskoufis
- 08-23 Dealing with endogenous regressors using copulas; on the problem of near multicollinearity, Dimitris Christopoulos, Dimitris Smyrnakis and Elias Tzavalis
- 09-23 A machine learning approach to construct quarterly data on intangible investment for Eurozone, Angelos Alexopoulos and Petros Varthalitis
- **10-23** Asymmetries in Post-War Monetary Arrangements in Europe: From Bretton Woods to the Euro Area, George Alogoskoufis, Konstantinos Gravas and Laurent Jacque
- **11-23** Unanticipated Inflation, Unemployment Persistence and the New Keynesian Phillips Curve, George Alogoskoufis and Stelios Giannoulakis
- 12-23 Threshold Endogeneity in Threshold VARs: An Application to Monetary State Dependence, Dimitris Christopoulos, Peter McAdam and Elias Tzavalis
- 13-23 A DSGE Model for the European Unemployment Persistence, Konstantinos Giakas
- 14-23 Binary public decisions with a status quo: undominated mechanisms without coercion, Efthymios Athanasiou and Giacomo Valletta
- 15-23 Does Agents' learning explain deviations in the Euro Area between the Core and the Periphery? George Economides, Konstantinos Mavrigiannakis and Vanghelis Vassilatos
- 16-23 Mild Explocivity, Persistent Homology and Cryptocurrencies' Bubbles: An Empirical Exercise, Stelios Arvanitis and Michalis Detsis
- 17-23 A network and machine learning approach to detect Value Added Tax fraud, Angelos Alexopoulos, Petros Dellaportas, Stanley Gyoshev, Christos Kotsogiannis, Sofia C. Olhede, Trifon Pavkov
- 18-23 Time Varying Three Pass Regression Filter, Yiannis Dendramis, George Kapetanios, Massimiliano Marcellino

<u>2023</u>



ATHENS UNIVERSITY



Department of Economics Athens University of Economics and Business

The Department is the oldest Department of Economics in Greece with a pioneering role in organising postgraduate studies in Economics since 1978. Its priority has always been to bring together highly qualified academics and top quality students. Faculty members specialize in a wide range of topics in economics, with teaching and research experience in world-class universities and publications in top academic journals.

The Department constantly strives to maintain its high level of research and teaching standards. It covers a wide range of economic studies in micro-and macroeconomic analysis, banking and finance, public and monetary economics, international and rural economics, labour economics, industrial organization and strategy, economics of the environment and natural resources, economic history and relevant quantitative tools of mathematics, statistics and econometrics.

Its undergraduate program attracts high quality students who, after successful completion of their studies, have excellent prospects for employment in the private and public sector, including areas such as business, banking, finance and advisory services. Also, graduates of the program have solid foundations in economics and related tools and are regularly admitted to top graduate programs internationally. Three specializations are offered:1. Economic Theory and Policy, 2. Business Economics and Finance and 3. International and European Economics. The postgraduate programs of the Department (M.Sc and Ph.D) are highly regarded and attract a large number of quality candidates every year.

For more information:

https://www.dept.aueb.gr/en/econ/